\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{https://ijrps.com}{\color{blue}\underline{\smash{https://ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\begin{document}


\def\RunningAuthor{Tamilselvi S and Nalini S J}
\firstPage{2674}
\articleType{Original Article}
\receivedDate{15.04.2019}
\acceptedDate{13.07.2019}
\revisedDate{08.07.2019}
\journalVolume{10}
\journalIssue{4}
\journalDoi{ijrps.v10i4.1526}
\copyrightYear{2019}
\def\authorCount{2}
\def\affCount{2}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{The prevalence of poly cystic ovary syndrome among young adult girls}
\author{Tamilselvi~S\textsuperscript{*}\textsuperscript{1}, 
            Nalini~S~J\textsuperscript{2}~\\[5pt]{\textsuperscript{1}Department of Community Health Nursing, Saveetha Institute of Medical and Technical Sciences, Thandalam, Chennai, Tamil Nadu, India}~\\{\textsuperscript{2}Faculty of Nursing, Sri Ramachandra Institute of Higher Education and Research, Chennai, Tamil Nadu, India}}

\begin{abstract}
Polycystic ovarian disorder (PCOS) is a typical endocrine issue experienced in ladies and is related with issues, for example, menstrual inconsistencies; anovulation; barrenness; hirsuitism; heftiness; insulin opposition, dyslipoproteinemia; and, in later life, with diabetes mellitus and uterine malignant growth. Early determination is fundamental for early treatment and to limit the prompt and constant results. The predominance of PCOS in our general public has not yet been resolved. This investigation is an endeavor to evaluate its pervasiveness of youthful grown-up young ladies 18-25 Years. PCOS was analyzed by utilizing self-appraisal apparatus and BMI, and the predominance was observed to be \% of the population.
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Young adult girls,\newline BMI,\newline PCOS
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Tamilselvi~S~\\ Phone:\ +91 99405 29561~\\  Email:\ tamildani@gmail.com
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{https://ijrps.com}{https://ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
Polycystic ovary disorder (PCOS) is a typical endocrine issue experienced in ladies of the regenerative age the world over. It was first depicted in 1935 by Stein and Levinthal and was known as the Stein{\textendash}Leviathan disorder. It is a heterogeneous issue with a familial premise. It is related with a wide range of introducing highlights, including anovulation, fruitlessness, corpulence, hirsuitism, insulin opposition, and dyslipoproteinemia. In roughly 20\% of asymptomatic cases were found on ultrasound assessment \unskip~\citep{586827:13603675}.

Pervasiveness of PCOS is expanding among Indians. PCOS in vagrant Indians have been assessed at 52\% level. The commonness of PCOS changes somewhere in the range of 2.5 and 11 \%. Late European and American examinations utilizing NIH criteria are in understanding that PCOS is a typical endocrine issue, influencing ladies of regenerative age up to 6.8\%. The introduction of PCOS in puberty is additionally influenced by metabolic changes related with muscle to fat ratio dissemination. Ordinary pubescence and youthfulness are related with expanded insulin opposition and expanded fasting insulin fixations\unskip~\citep{586827:13603677}. Ladies with PCOS, regardless of whether lean or large, have higher insulin obstruction than ladies without PCOS of comparable weight. Besides, thinks about show exacerbating of insulin opposition with overweight and heftiness related with PCOS Pelusi, 2002 \unskip~\citep{586827:13603676}.
    
\section{Materials and Methods}

\begin{table*}[!htbp]
\caption{\boldmath {Self-Assessment Tool Questionnaire} }
\label{tw-f01464cb937f}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.06779999999999999\linewidth-2\tabcolsep}p{\dimexpr.4633\linewidth-2\tabcolsep}p{\dimexpr.11610000000000003\linewidth-2\tabcolsep}p{\dimexpr.11850000000000003\linewidth-2\tabcolsep}p{\dimexpr.1185\linewidth-2\tabcolsep}p{\dimexpr.1158\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}{\multirow{2}{*}{S.n}} & {\multirow{2}{*}{Clinical signs and symptoms of PCOS}} & \multicolumn{2}{p{\dimexpr(.23460000000000008\linewidth-2\tabcolsep)}}{\cAlignHack Yes} & \multicolumn{2}{p{\dimexpr(.2343\linewidth-2\tabcolsep)}}{\cAlignHack NO}\\
\rowcolor{kwdboxcolor} &  & N &  \% & N &  \%\\
\tblmidrule 
1. &
  Do you have 8 or few periods a year? &
  281 &
  52.5\% &
  254 &
  47.5\%\\
2 &
  Have you ever gone 4 months or longer without having a period? &
  129 &
  24.1\% &
  406 &
  75.9\%\\
3 &
  Do you have irregular bleeding or spotting &
  114 &
  21.3\% &
  421 &
  78.7\%\\
4 &
  Do you have excessive hair on your face or body &
  203 &
  37.9\% &
  332 &
  62.1\%\\
5 &
  Do you have a severe adolescent or persistent adult acne &
  73 &
  13.6\% &
  462 &
  86.4\%\\
6 &
  Do you have thinning scalp hairs &
  357 &
  66.7\% &
  178 &
  33.3\%\\
7 &
  Do you have skin tags or velvety dark skin patches on the nape of your neck &
  46 &
  8.6\% &
  489 &
  91.4\%\\
8 &
  Have you recently had a significant weight gain &
  127 &
  23.7\% &
  408 &
  76.3\%\\
9 &
  Do you carry excessive weight around your waist line &
  182 &
  34.0\% &
  353 &
  66.0\%\\
10 &
  Do you feel sugar craving, drowsiness or sometimes light headedness within the first few hours after a meal? &
  66 &
  12.3\% &
  469 &
  87.7\%\\
11 &
  Do you or any close family members have type 2 diabetes mellitus &
  176 &
  32.9\% &
  359 &
  67.1\%\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {BMI - Prevalence Data} }
\label{tw-b3c19754b44b}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.24139999999999997\linewidth-2\tabcolsep}p{\dimexpr.21710000000000004\linewidth-2\tabcolsep}p{\dimexpr.25169999999999998\linewidth-2\tabcolsep}p{\dimexpr.2898\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor} & Size & Mean\ensuremath{\pm}SD & 95\% CI\\
\tblmidrule 
BMI &
  535 &
  22.12 \ensuremath{\pm} 2.593 &
  24.034\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Self Assessment Tool - Prevalence Data} }
\label{tw-6f192052a599}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.2396\linewidth-2\tabcolsep}p{\dimexpr.2223\linewidth-2\tabcolsep}p{\dimexpr.25349999999999998\linewidth-2\tabcolsep}p{\dimexpr.2846\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor} & Size & Mean\ensuremath{\pm}SD & 95\% CI\\
\tblmidrule 
Self-assessment  \mbox{}\protect\newline Tool &
  535 &
  3.28 \ensuremath{\pm} 2.44 &
  3.28\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}
The study was conducted to estimate the prevalence of PCOS among the female students of Saveetha institute of medical and technical sciences. Descriptive survey method adapted for this study. The sampling technique was convenience sampling. SIMATS consists of eight constituent colleges with various courses and equipped with 1500 bedded multi-super speciality hospital. Out of this, a researcher has selected two colleges like physiotherapy and engineering. Students were gathered in the conference hall, investigator along with student's co-coordinator from respective colleges gave instructions about PCOS. Questionnaire was distributed to the participants, and informed consent was obtained around 535 young adult girls were participated in this prevalence study, by using convenience sampling technique. Self-assessment tool included history about their weight gain, puberty, menstrual pattern, presence of hirsuitism, dysmenorrhoea, persistent acne, past and present medical and surgical history was obtained. It was interpreted, and abnormality was noted. 

\textbf{Scoring and interpretation }

Self-assessment tool was used based on (Rotterdam criteria) It consisted of eleven questions regarding the signs and symptoms of PCOS, Total items were 11 with a total score of 11 the tool was interpreted as no problem (0) and yes means (1) from the total score 11 those who scored 5 and above were considered as suspected with PCOS. Body mass index of up to 22.9kg/m\ensuremath{^{2}} comes under as normal, 23 to 24.9kg/m\ensuremath{^{2}} comes under as overweight, and more than 25kg/m\ensuremath{^{2}} were considered as obese according to the WHO (world health organization). 
    
\section{Results and Discussion}
Based on the BMI  values among the participants in prevalence study the mean value is 22.122 with 0.114 SE with the C.I of mean 0.224, the result shows about 24.03\% (95\% CI) were found symptoms with PCOS among 535 young adult girls. The mean value of self-assessment tool was 3.279 with 0.106 SE. The result shows about 29.82 (95\% CI) were found symptoms with PCOS among 535 young adult girls. Data analysis and interpretation by descriptive statistics.

PCOS among young adult is a raising problem in India which needs careful assessment, timely intervention, and early treatment. The diverse manifestations of PCOS start at an age when a female is maturing into an adult. It is essential to make an early diagnosis in order to  prevent early PCOS syndrome and its complications. Hence, the present study was conducted to find the prevalence of PCOS among young adult girls and assess the effectiveness of self-help strategies among the students studying in Saveetha Institute of Medical and technical sciences. The present study reveals that the prevalence of PCOS was found to be 24.034\% (BMI) and 4.00\% (self-assessment tool). 

All around, pervasiveness evaluations of PCOS are exceptionally factor, extending from 2.2\% to as high as 26\%. In India, the pervasiveness is bit by bit expanding. As per an examination directed by \unskip~\citep{586827:13603678} on 460 young ladies matured 15-18 years in a private school in Andhra Pradesh, South India, the commonness of PCOS was observed to be 9.13\% in youths. In comparable examination led on Saudi young ladies, the assessed pervasiveness of PCOS was seen to be 53.7\% which is higher. Vidya R . found that as per the World Health Organization, there are PCOS influenced 116 million ladies worldwide in 2012 (3.4\%) of ladies. 

A Similar discoveries were gotten in a study led that the greater part of the building understudies had lacking learning in a pre-test. Another examination done to gauge the impact of instructive sessions about Polycystic Ovarian Syndrome in late youthful young ladies indicated the same outcomes, Hanan Elsayed Mohamed, 2015.  It was found in a concentrate that a large number of members were suspected with PCOS as they had signs and side effects. It demonstrates that numerous females don't counsel gynecologists to analyze their wellbeing status. It was found in another examination that with the exception of in hazardous circumstances, the families don't take the lady to a medical clinic, since removing a lady from their homes had thought to be ill-bred Rizvi, Khan, and Shaikh, 2014. 

An examination done among young people in Karachi found that the vast majority of the subjects expressed that the issues identified with regenerative issues ought not to be depicted to anybody. Just 22\% revealed that the issue ought to be talked about with the relatives, though remaining subjects expressed that if there should arise an occurrence of any issue, a specialist ought to be counseled Ali, Ali, Waheed, and Memon, 2006. It is accepted that spearmint and exercise positively affect BMI, Improvement in pre-adult young ladies with PCOS Trent, 2005. The present commonness study Among the 535 subjects, 83 had the indication of PCOS score of over 5 in the statistic part demonstrated the youthful grown-up young ladies lion's share that is 208 (38.9\%) of youthful grown-up young ladies were in the age bunch between 18-19 years. In a dietary example, the greater part of them are has a place with non-vegan 432 (80.7\%) (Tables~\ref{tw-f01464cb937f} and~\ref{tw-b3c19754b44b} \&Table~\ref{tw-6f192052a599}).
    
\section{Conclusion}
The finding of the study gives that commonness of indications and manifestations of PCOS are expanding. However, understudies didn't know about PCOS in spite of the fact that its signs and side effects were knowledgeable about a significant number of them. So extraordinary mindfulness projects ought to be done to give learning about such sicknesses to females. Besides, the study demonstrates that the greater part of females don't counsel gynecologist except if there is a serious or hazardous issue or sickness. So females ought to counsel gynecologist at any rate once in a year to analyze their wellbeing status to improve the wellbeing. 
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
